**Author’s response to reviews**

**Title:** Thai medical students' attitudes regarding what constitutes a "good death": a multi-center study

**Authors:**

Panita Limpawattana (lpanit@kku.ac.th)

Varalak Srinonprasert (varalaksi@gmail.com)

Machumad Manjavong (manchumad@hotmail.co.th)

Srivieng Pairojkul (srivieng@kku.ac.th)

Jarin Chindaprasirt (jarich@kku.ac.th)

Sawadee Kaiyakit (talasan247@gmail.com)

Thitikorn Juntararuangtong (thitikornJuntararuangtong@gmail.com)

Kongpob Yongrattanakit (kongyo@kku.ac.th)

Thunchanok Kuichanuan (thunchanok.kuichanuan@gmail.com)

**Version:** 2  **Date:** 27 Feb 2019

**Author’s response to reviews:**

1. Please clarify whether the questionnaire used in your study was developed for this study or has previously been published elsewhere. If the questionnaire has been published elsewhere please provide a reference to it in your manuscript, if the questionnaire was developed for this study please upload an English language version as a supplementary file.


2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
Response: all necessary changes have been made

Thank you for considering our article for publication